Author’s response to reviews

Title: The adaptation and evaluation of a CBT-based manual for the inpatient treatment of youth depression: a pilot study

Authors:

Michael Frey (Michael.frey@med.uni-muenchen.de)
Kathrin Pietsch (kathrin.pietsch@med.uni-muenchen.de)
Laura Weninger (Laura.weninger@med.uni-muenchen.de)
Matthias Brockhaus (brockhausmatthias@gmail.com)
Fabian Loy (Fabian.Loy@puk.zh.ch)
Nathalie Claus (nathalie.claus@online.de)
Petra Wagenbüchler (petra.wagenbuechler@med.uni-muenchen.de)
Selina Kornbichler (selina.kornbichler@gmx.de)
Jana Kroboth (Jana.Kroboth@med.uni-muenchen.de)
Beatrice Georgii (beatrice.georgii@cip-tagesklinik.de)
Inga Wermuth (Inga.wermuth@med.uni-muenchen.de)
Anna Kititsina (Anna.Kititsina@kbo.de)
Katharina Heiser (Katharina.heiser@med.uni-muenchen.de)
Gerd Schulte-Koerne (Gerd.schulte-koerne@med.uni-muenchen.de)
Belinda Platt (belinda.platt@med.uni-muenchen.de)

Version: 2 Date: 27 Jan 2020

Author’s response to reviews:

Dear Gillian Lancaster and Lehana Thabane,

Manuscript: The adaptation and evaluation of a CBT-based manual for the inpatient treatment of adolescent depression: a pilot study

We would like to thank the Editors and the Reviewers for taking time to review our manuscript revisions. We have made the following changes as recommended by the editor:
1) Abstract: Please report estimates of effect (95% CI). Consider deleting p-values since the study was not designed to address statistical significance and therefore these may represent possible false-positive.

Thank you for this suggestion, we have adopted it in the submitted manuscript on p. 2.

2) Results: Again, please report estimates of effect along with their 95% CI, and consider deleting the p-values.

Thank you for this suggestion, we have adopted it in the submitted manuscript on p. 11 and p.12.

Yours sincerely,

Belinda Platt (corresponding author)

Department of Child and Adolescent Psychiatry, Psychosomatics and Psychotherapy
University Hospital Munich, Email: Belinda.platt@med.uni-muenchen.de